 vc/21-Jul-05
Week 29  starting 18 July 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  17 July 05 @: 7/18/2005 1:08:55 PM
	3
	A1
	MBBL2189
	Prep 5 Cool Down
	102.35

	6
	A2
	MBAL2051
	Prep 5 Cool Down
	260.84

	7
	B1
	SSS199
	ICS 4 Final connection
	300.46

	10
	B2
	SSS036
	Prep 2.2 IAP @ warm HF
	301.17

	5
	C1
	MBAR3369
	Prep 5 Cool Down
	289.19

	4
	C2
	MBBL3085
	PT 13.1 IAP @ Warm
	299.11

	2
	D1
	SSS217
	PT 12 Warm Up
	100.57

	1
	D2
	SSS617
	Prep 5 Cool Down
	3.73 HV

	9
	E1
	MBAL3109
	ICS 4 Final connection
	301.35

	11
	E2
	MBAR3373
	PT 14 Magnet MAPS
	301.89

	12
	F1
	MBBL2206
	PT 14 Magnet MAPS
	300.55

	8
	F2
	MBBL3126
	Prep 5 Cool Down
	300.82


   Temperature data are taken from TBSM18 site

 
Coordinator this week: Marco;MTM support 16-3602
14 complete Magnet Tests cycles   (no ROBTC this week )
11 dipoles( 2 with MM )+ 2 SSS( 1  with MM & SSW ) +  1 SSSS 
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· General Points  :
· Special SSSS (Jacques)

· Special Tests D1: 217

· Special Tests B1 : Connection cryostat(CC)

· AOB
Tests 

	Magnets under Test from 10-07-2005 to 16-07-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR3365 
	[image: image1.png]



	


	08-07-2005 
	26 h.
	19 h.
	25 h.
	16 h.
	12 h.
	12-07-2005 
	98 h.
	Stripping
	
	

	2
	TBA2
	MBAR3306 
	[image: image3.png]



	


	11-07-2005 
	30 h.
	25 h.
	16 h.
	16 h.
	16 h.
	15-07-2005 
	103 h.
	Stripping
	
	

	3
	TBB1
	SSS174 
	[image: image5.png]



	


	10-07-2005 
	31 h.
	15 h.
	13 h.
	11 h.
	13 h.
	13-07-2005 
	83 h.
	Stripping
	
	

	4
	TBC1
	MBAL1245 
	[image: image7.png]



	


	11-07-2005 
	39 h.
	26 h.
	18 h.
	12 h.
	21 h.
	16-07-2005 
	116 h.
	Stripping
	
	

	5
	TBC1
	MBBL2197 
	[image: image9.png]



	


	07-07-2005 
	27 h.
	23 h.
	20 h.
	14 h.
	15 h.
	11-07-2005 
	99 h.
	Stand By
	
	

	6
	TBC2
	MBBL3156 
	[image: image11.png]



	


	09-07-2005 
	37 h.
	31 h.
	34 h.
	15 h.
	12 h.
	15-07-2005 
	129 h.
	Stripping
	
	

	7
	TBD1
	SSS203 
	[image: image13.png]



	


	29-06-2005 
	46 h.
	42 h.
	126 h.
	10 h.
	41 h.
	10-07-2005 
	265 h.
	Stripping
	
	

	8
	TBD2
	SSS643 
	[image: image15.png]



	


	11-07-2005 
	32 h.
	12 h.
	7 h.
	16 h.
	26 h.
	15-07-2005 
	93 h.
	Stand By
	
	

	9
	TBE1
	MBAR3323 
	[image: image17.png]



	


	05-07-2005 
	45 h.
	34 h.
	15 h.
	14 h.
	18 h.
	10-07-2005 
	126 h.
	Stripping
	
	

	10
	TBE1
	MBBR3292 
	[image: image19.png]



	


	11-07-2005 
	31 h.
	30 h.
	26 h.
	17 h.
	18 h.
	16-07-2005 
	122 h.
	Stripping
	
	

	11
	TBE2
	MBAL1244 
	[image: image21.png]



	


	09-07-2005 
	31 h.
	24 h.
	17 h.
	15 h.
	20 h.
	13-07-2005 
	107 h.
	Stand By
	
	

	12
	TBF1
	MBAL3114 
	[image: image23.png]



	


	09-07-2005 
	33 h.
	27 h.
	17 h.
	16 h.
	9 h.
	13-07-2005 
	102 h.
	Stripping
	
	

	13
	TBF2
	MBAR3349 
	[image: image25.png]



	


	10-07-2005 
	30 h.
	23 h.
	52 h.
	12 h.
	17 h.
	16-07-2005 
	134 h.
	Stripping
	
	

	14
	TBF2
	MBBL1238 
	[image: image27.png]



	


	05-07-2005 
	39 h.
	25 h.
	14 h.
	14 h.
	18 h.
	10-07-2005 
	110 h.
	Stripping
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR3365
	10720 A.
(D2-U-S01)
	12652 A.
(D1_L_S13)
	12850 A.
[image: image29.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR3306
	8780 A.
(D1-U)
	12713 A.
(D1-L)
	12850 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS174
	11834.1 A.
(QE_34)
	12230 A.
(QE-12)
	12398.9 A.
(QI-34)
	[image: image31.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAL1245
	12375 A.
(D1-L)
	12355.1 A.
(D2_U)
	12772 A.
(D2-L)
	[image: image32.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL2197
	10396 A.
(D2-L)
	10895 A.
(D1-U)
	12542 A.
(D1-U)
	12850 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL3156
	9947 A.
(D2-U)
	11381 A.
(D1-U)
	11937 A.
(D2 - U)
	12087 A.
(D2 - U)
	12409 A.
(D2 - L)
	12624.2 A.
(D2-U)
	12755 A.
(D1_L)
	12851 A.
(D2_L)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-

	7
	SSS203
	12068 A.
(QE_34)
	12318 A.
(QE_12)
	12421 A.
(QI_34)
	[image: image35.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS643
	5650 A.
(MQ_I_12)
	5820 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBAR3323
	12271 A.
(D1-L)
	12668 A.
(D1-L)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBR3292
	12591 A.
(D1-U)
	12850 A.
[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL1244
	11779.6 A.
(D1_L)
	12495 A.
(D1-L)
	12850 A.
(D2-L)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAL3114
	10041 A.
(D2-U)
	12330 A.
(D1-U)
	12850 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAR3349
	0 A.
(No Data)
	10226 A.
(D2_U_S13)
	10992 A.
(D1-L-S01)
	12848.5 A.
(D1_L_S01)
	12850 A.
[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBL1238
	10903.9 A.
(D1_U)
	12493.4 A.
(D1_L)
	12459 A.
(D1_L)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	
	
	
	
	
	
	
	
	
	
	
	
	
	
	


SSW was done at warm , at  cold & again at warm on SSS203 on D1 = 265 hrs  total 
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SM18 Operation issues – July 2005, Thu 14 to Tue 19

	Date
	TB
	Magnet
	Team
	Problem
	Solution

	Wed 13, 01:56
	C2
	3156
	AB/CO
	Server not responding
	Sun reset by MTM op (Florence ?)

	Thu 14, 10:00
	
	
	AT/ACR
	Cryo is slow
	No solution. First priority given to SSS217 (TBD1) for long-term fatigue test.

	Thu 14, 16:25
	C1
	1245
	AT/MTM
	Q during SPA @ 1672A
	Marco suggests to repeat test (magnet safety not in danger)

	Fri 15,  09:20
	B2
	204
	AT/MTM
	1.3V offset in Q-I-E
	Marylin investigates, no result ?

	Fri 15,  10:09
	D1
	217
	AT/MTM
	Nominal current not reached during EF corrector powering
	Eric switched to the right configuration.

	Fri 15, 19:58
	A1
	2189
	AT/MTM
	(Markus)
	Richard corrects wrong gains of Potaim card 3, Ch. 1

	Fri 15, 22:20
	D1
	217
	AT/MTM
	SPA after 9 minutes @ Iultimate
	No explanation found, probable cryo instability (test repeated) 

	Sat 16, 02:20
	B2
	204
	AT/MTM
	Q during SPA @ 11410A. Test repeated, and the Q occurred a second time. 
	Richard investigates and finds a probable short on a parallel protection resistor. The trigger threshold is temporarily increased by a factor 3 to allow the SPA to proceed. Later warm measurements, however, are OK and do not confirm this hypothesis. Magnet put in standby.

	Sat 16, 08:30
	A1
	2189
	AT/MTM
	Q during SPA @ 10220A.
	Test was repeated successfully. 

	Sat 16, 13:28
	F1
	2206
	AT/MTM
	False trigger on HF. LF show error loading config file. (Guy ?)
	Unexplicable.

	Sat 16, 16:21
	E1
	3292
	ICS
	ROCLA arrives early on the bench (same as on A2 bench on Fri)
	Is this a problem ?

	Sat 16, 16:40
	A1
	2189 (?)
	AT/MTM
	Q during SPA @ 8293A.
	Unexpla

	Sun 17, 13:56
	C2
	3085
	AT/MTM
	Q in the busbars @ PT5.2.
	Richard finds a very symmetric quench, which is detected only outside of the magnet. Go on with the test.

	Sun 17, 23:37
	B2
	036
	ICS
	Connections appear to have been re-labeled.
	Re-labeling is correct (sick magnet).

	Sun 17, 04:56
	B2
	??
	AT/MTM
	Unable to launch IAP final due to cryo interlock. (Eric ?)
	OK after TEMA reset.











